Author's response to reviews

Title: Consequences of spinal pain: Do age and gender matter? A Danish cross-sectional population-based study of 34,902 individuals 20-71 years of age.

Authors:

Charlotte Leboeuf-Yde (clyde@health.sdu.dk)
René Fejer (rene.fejer@slb.regionsyddanmark.dk)
Jan Nielsen (jan.nielsen2@ouh.regionsyddanmark.dk)
Kirsten O Kyvik (kkyvik@health.sdu.dk)
Jan Hartvigsen (jhartvigsen@health.sdu.dk)

Version: 5 Date: 31 January 2011

Author's response to reviews:

Dear Editors,

Further formatting changes have been made according to the revisions put forth by the editors.

We apologies for any inconvenience caused.

Sincerely,

René Fejer